Author's response to reviews

Title: Protocol for the implementation and evaluation of a community-based intervention seeking to reduce dietary salt intake in Lithgow, Australia

Authors:

Mary-Anne Land (maland@georgeinstitute.org.au)
Paul Jeffery (pje@deakin.edu.au)
Jacqui Webster (jwebster@georgeinstitute.org.au)
Michelle Crino (mcrino@georgeinstitute.org.au)
John Chalmers (chalmers@georgeinstitute.org.au)
Mark Woodward (markw@georgeinstitute.org.au)
Caryl Nowson (caryl.nowson@deakin.edu.au)
Wayne Smith (wsmith@doh.health.nsw.gov.au)
Victoria Flood (vicki.flood@sydney.edu.au)
Bruce Neal (bneal@georgeinstitute.org.au)

Version: 2
Date: 5 February 2014

Author's response to reviews: see over
5 February 2014

Ms Natalie Pafitis
Floor 6, 236 Gray's Inn Road
London
WC1X 8HB
United Kingdom

Dear Editor-in-Chief

Article for submission:

“Protocol for the implementation and evaluation of a community-based intervention seeking to reduce dietary salt intake in Lithgow, Australia”

Thank you for your consideration of our manuscript for publication in BMC Public Health.

Please find below our responses to the points raised in email communication dated 05.02.2014.

1. Ethical and Funding Approval Documentation - we have emailed copies of all ethical approval and funding approval to BMCSeriesEditorial@biomedcentral.com

2. Funding – the study is funded by an Australian National Health and Medical Research Council grant

3. Study status - the study is ongoing, the follow up data collection is scheduled for March-April 2014 with analysis to commence shortly after completion.


5. Please format your authors contribution section – thank you we have revised this section in line with the suggested instructions and re-submitted the manuscript.

6. Please format your Competing Interest section – we have included our competing interest section.

7. Please confirm that your clinical trial is in a publicly accessible registry before we
begin peer review - not application.

Kind Regards

Mary-Anne Land
T +61 2 9993 4547 | F +61 2 9993 4502
E maland@georgeinstitute.org.au